Author's response to reviews

Title: Concealed Primary Aortic Sarcoma induced Hypertensive Encephalopathy resulting from a Thoracic Aortic Occlusion: a case report

Authors:

Hyunmin Choi (hmchoi49@naver.com)
Woo-Ik Jang (cwi87@paik.ac.kr)
Chang-Young Kim (lion0052@paik.ac.kr)
Jae-Jin Kwak (jjkwak@paik.ac.kr)
June Namgung (jinamgung@paik.ac.kr)
Joon-Hyung Doh (djih95@paik.ac.kr)
Sung-Yun Lee (im2pci@paik.ac.kr)

Version: 3 Date: 16 April 2013

Author's response to reviews: see over
Author's response to editorial recommendations

Title: Concealed Primary Aortic Sarcoma induced Hypertensive Encephalopathy resulting from a Thoracic Aortic Occlusion: a case report

Authors:

Hyunmin Choi (hmchoi49@naver.com),

Hee-Jeoung Yoon (kestus@hanmail.net): We added her in this revision.

Woo-Ik Jang (cwi87@paik.ac.kr),

Joon-Hyung Doh (djh95@paik.ac.kr)

Version: 2 Date: 16 April 2013

Author's response to recommendations: see over
The Biomed Central Editorial Team

Object MS: 1406864839847652 - Concealed Primary Aortic Sarcoma Provoked Hypertensive Encephalopathy due to Thoracic Aortic Occlusion: a case report.
Dr Hyunmin Choi et al.

1. Please complete the description of the patient’s relevant demographic information (without adding any details that could lead to the identification of the patient, eg. Caucasian, Japanese, African-American, etc.)

We described the patient’s demographic information.

2. Please include a list of abbreviations used in the manuscript and their meanings

We made a list of abbreviations.

3. Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements

We made acknowledgement section and disclosed the source of funding.